Author's response to reviews

Title: Cardiac damage after treatment of childhood cancer: A long-term follow-up

Authors:

Veronika Velensek (veronika.velensek@kclj.si)
Uros Mazic (uros.mazic@mf.uni-lj.si)
Ciril Krzisnik (ciril.krzisnik@mf.uni-lj.si)
Damjan Demsar (damjan.demsar@ijs.si)
Janez Jazbec (janez.jazbec@mf.uni-lj.si)
Berta Jereb (bjereb@onko-i.si)

Version: 5 Date: 15 April 2008

Author's response to reviews:

Dear The BioMed Central Editorial Production Team.

we are re-submitting revised version of our manuscript. All revisions were done according to your suggestions. Specifically we have:
1.removed the authors' qualifications,
2.we have removed Keywords, Word-counts, Running head and Running title
3.we have removed the title from the head of the 'Abstract' page
4.Introduction - this section was renamed `Background
5. Figures and tables were handeled as you requested.
If we have missed anything please let us know.

Sincerely,

Veronika velenšek